Child Labor, Schooling, and Child Ability*

Richard Akresh University of Illinois at Urbana-Champaign Emilie Bagby Mathematica Policy Research

Damien de Walque The World Bank

Harounan Kazianga Oklahoma State University

September 29, 2012

Abstract

Using data we collected in rural Burkina Faso, we examine how children's cognitive abilities influence households' decisions to invest in their education. To address the endogeneity of child ability measures, we use rainfall shocks experienced in utero or early childhood to instrument for ability. Negative shocks in utero lead to 0.24 standard deviations lower ability z-scores, corresponding with a 38 percent enrollment drop and a 49 percent increase in child labor hours compared with their siblings. Negative education impacts are largest for in utero shocks, diminished for shocks before age two, and have no impact for shocks after age two. We link the fetal origins hypothesis and sibling rivalry literatures by showing that shocks experienced in utero not only have direct negative impacts on the child's cognitive ability (fetal origins hypothesis) but also negatively impact the child through the effects on sibling rivalry resulting from the cognitive differences.

Keywords: Fetal origins hypothesis; Education; Child Labor; Sibling Rivalry *JEL classification*: I21, J12, O15, J13

Contact Information: Richard Akresh, University of Illinois at Urbana-Champaign, Department of Economics, 1407 West Gregory Drive, Room 101, Urbana, IL 61801. Email: akresh@illinois.edu;

Emilie Bagby, Mathematica Policy Research, 1100 1st Street, N.E., 12th Floor, Washington, D.C., 20002. Email: ebagby@mathematica-mpr.com;

Damien de Walque, The World Bank, Development Research Group, 1818 H Street, N.W., Washington, D.C., 20433. Email: ddewalque@worldbank.org;

Harounan Kazianga, Oklahoma State University, Department of Economics, 324 Business Building, Stillwater, OK 74708. Email: harounan.kazianga@okstate.edu

_

^{*} The data used in this paper were collected for a project evaluating social protection programs in Burkina Faso, which benefited from support by Marie-Claire Damiba, Seydou Kabré, and Victorine Yameogo from the Secrétariat Permanent du Comité National de Lutte contre le SIDA et les Infections Sexuellement Transmissibles (SP-CNLS-IST) in Burkina Faso and Hans Binswanger, Nono Ayivi-Guedehoussou, Ousmane Haidara, Timothy Johnston, Mead Over, and Tshiya Subayi-Cuppen at the World Bank. Robert Ouedraogo, Jean-Pierre Sawadogo, Bambio Yiriyibin and Pam Zahonogo from the University of Ouagadougou, Department of Economics supervised the data collection. The project is funded by the NBER Africa Project and the following World Bank trust funds grants: Spanish Strategic Impact Evaluation Fund (SIEF), Bank-Netherlands Partnership Program (BNPP), Gender Action Plan (GAP), Knowledge for Change Program (KCP), WB-DFID Evaluation of the Community Response to HIV and AIDS, and Luxembourg Poverty Reduction Partnership (LPRP). The authors also thank German Caruso and Moussa Kone for research assistance and Christopher Udry, Aziz Ouedraogo, Zah Marie Sawadogo, Dembélé Youssouf, and Samuel Alnaa for assistance procuring the rainfall data. We are grateful to Adam Wagstaff for useful comments. The findings, interpretations, and conclusions expressed in this paper are entirely those of the authors. They do not necessarily represent the views of the World Bank, its Executive Directors, or the countries they represent.

1. Introduction

Based on 2007 data, only 76 percent of primary school-aged children were enrolled in sub-Saharan Africa, which is the lowest net enrollment rate of all developing country regions (United Nations 2010). The estimated 31 million not enrolled African children represent over 45 percent of the world's total of out-of-school children. Over half of the countries in sub-Saharan Africa do not provide legal guarantees for free primary school education. In settings without universal free primary education and where most children engage in child labor activities, as is true in many African countries, understanding how parents decide how to allocate limited resources across siblings is important as these decisions have long-term impacts on each child's future earnings, health, and overall welfare. Schooling decisions depend on parent perceptions about the returns to school for a child, the opportunity cost of sending that child to school, the child's ability, and the presence and characteristics of the child's siblings. To raise levels of school enrollment, it is critical to understand these links between child ability, school enrollment, and child labor.

An extensive literature examines the source of inequalities for children's educational investments within a household building on seminal work by Becker and Tomes (1976) that highlights the tradeoff between child quantity and their 'quality.' In making decisions about schooling and child labor, parents have information about their children's abilities that is often not available to researchers, a fact that partly explains why much of the empirical research on the determinants of parental investments in children focuses on easy-to-observe demographic

¹ This compares with an average net enrollment rate of 89 percent in developing regions and 96 percent in developed regions. These enrollment rates in Africa still represent an enormous improvement since 1998 when African net enrollment rates were only 58 percent. In Burkina Faso, the focus of this paper, the net enrollment rate was 35 percent in 2000 and improved to 55 percent by 2007 (UN Statistics Division). Based on this paper's data from Burkina Faso, only 24 percent of households with multiple primary school aged children enroll all of their children. ² Based on the UNICEF Multiple Indicator Cluster Survey End of Decade Assessment, almost three-fourths of all rural primary school-aged children in developing countries were involved with some type of work that included helping with the family farm or business or providing domestic household services such as cooking, sweeping, doing dishes, or fetching water (Edmonds and Pavcnik 2005).

characteristics of the child such as gender, birth order, or family composition (Parish and Willis 1993; Garg and Morduch 1998; Black, Devereux, and Salvanes 2005; Emerson and Souza 2008).³ More recent papers attempt to use direct measurements of a child's ability such as IQ scores (Kim 2005), achievement tests (Glick and Sahn 2010), or cognitive tests (Ayalew 2005) to better understand which factors influence investment decisions.

The purpose of this paper is to examine the role that a child's cognitive ability plays in a household's decision to invest in that child's education or to use the child for labor activities. However, the difficulty with using cognitive ability measured in a survey at the same time as outcomes is that ability may reflect the accumulation of previous investments and parental preferences throughout the child's life. To address this potential endogeneity problem, we use an instrumental variables strategy in our within-household estimates of sibling rivalry that allows for a causal estimate of the impact of ability on both schooling and child labor decisions.

Specifically, we use rainfall shocks in the child's village that were experienced in utero or during early childhood to instrument for a child's cognitive ability. In a setting where most parents are rain-fed subsistence farmers, negative rainfall is an exogenous shock that reduces the harvest and amount of food and nutrition available over the following year. The fetal origins hypothesis, often associated with Barker (1998), argues that health and nutrition shocks suffered in utero can cause irreversible adaptations to the local food environment and that children cannot catch up even if they later have good nutrition and health care. These conditions experienced in utero or

³ See Strauss and Thomas (1995) and Glewwe and Kremer (2006) for reviews of the literature. Related research explores the relationship between these demographic characteristics and the non-schooling outcomes of employment (Kessler 1991) and risky behaviors (Aizer 2004). Research looking explicitly at child labor generally finds negative effects of work on human capital accumulation (see Boozer and Suri (2001) and Beegle, Dehejia, and Gatti (2009) for specific examples from Ghana and Vietnam and Edmonds (2008) for an extensive review of the child labor literature). While this negative schooling-child labor relationship is consistently found in the literature, it is sensitive to the definition of work (see Levison, Moe, and Knaul (2001) documenting that a tradeoff between schooling and work depends on whether work includes domestic work). For a discussion of the theoretical background on child labor decisions, see Basu (1999).

early in life have been shown to have persistent and long-term effects on health, education, and socioeconomic outcomes (see seminal work by Stein et al. (1975) and more recent papers by Alderman, Hoddinott, and Kinsey (2006), Almond (2006), and Akresh, Lucchetti, and Thirumurhy (2012) that study extreme shocks such as droughts, epidemics, or wars, or papers by Behrman and Rosenzweig (2004), Oreopoulos et al. (2008), and Royer (2009) that use twin comparisons). Knudsen (2004) highlights that different age periods during a child's early development are critical for cognitive development. Recent work focuses on positive shocks during these critical childhood ages and finds improvements in cognitive ability due to nutritional supplementation in Guatemala (Maluccio et al. 2009), iodine supplementation in Tanzania (Field, Robles, and Torero 2009), a de-worming intervention in Kenya (Ozier 2011), cash transfers in Nicaragua (Macours, Schady, Vakis 2012), and child health interventions in the Matlab study area of Bangladesh (Barham 2012).

This paper makes three main contributions to the literature. First, by using an instrumental variables strategy in our sibling rivalry estimations, this is the first paper that is able to present causal estimates of the impact of cognitive ability on education and child labor outcomes. We combine this instrumental variables approach with an empirical identification strategy that focuses on within-household estimates of the relationship between ability and schooling and labor outcomes; this allows us to control for household factors that are constant across siblings and propose a sibling ability rivalry explanation for our results. Furthermore, unlike previous sibling rivalry studies, our explanation does not depend on assumptions about gender differences in returns to education (Morduch 2000) or in home production (Edmonds

⁴ See Almond and Currie (2011) for a review of the fetal origins hypothesis literature, Strauss and Thomas (2008) for a review of the link between early childhood health and later life outcomes, and Grantham-McGregor, Fernald, and Sethuraman (1999) for a review of the literature measuring the short and long-term effects of malnutrition and micronutrient deficiencies. Despite much of the literature focusing on extreme shocks, Maccini and Yang (2009) find evidence in Indonesia of negative long-term impacts due to much smaller rainfall shocks.

2006). Second, our first stage estimates that provide strong evidence of the link between negative shocks experienced in utero or during early childhood on subsequent cognitive ability contribute to the fetal origins hypothesis literature by highlighting a potential mechanism, reduced cognitive ability, through which observed impacts on adult outcomes such as earnings and occupation can be explained. We are also able to estimate placebo regressions showing that rain shocks prior to when the child was in utero show no relationship with ability, further strengthening our interpretation of a causal relationship. Third, because of the unique household survey data collected by the authors, this paper is the first to examine the relationship between cognitive ability and child labor, focusing not only on total hours of child labor, but also changes in the specific tasks that each child performs.⁵

We find that higher ability children, compared to their lower ability siblings, are more likely to be enrolled in school and perform fewer hours of child labor. We also find the child labor they do is less concentrated in tasks requiring long, continuous blocks of time to complete. Rainfall more than one standard deviation below the village's long-run historical average that is experienced by a child in utero is correlated with 0.24 standard deviations lower cognitive ability z-scores and compared to their siblings, that corresponds to a 38 percent lower likelihood of enrollment and a 49 percent increase in the number of hours of child labor. The negative enrollment impacts are largest for children who experience a rain shock while in utero, and while

⁵ The current paper builds on previous research (Akresh et al., 2012), which also explores the relationship between a child's cognitive ability and educational investments, but there are three significant differences between the current paper and the earlier work. First, the econometric identification strategy in the previous work only allows for an examination of correlations between child ability and enrollment, while in this paper we are able to measure a causal impact of ability on schooling by implementing an instrumental variables approach using the rain shock instruments. Second, in contrast with the previous paper that focuses exclusively on sibling rivalry, the current paper links the literature on the fetal origins hypothesis with the literature on sibling rivalry by showing that shocks experienced in utero or early childhood not only have direct negative impacts on the child's cognitive ability (fetal origins hypothesis), but also negatively impact the child through the effects on sibling rivalry resulting from the cognitive differences. Third, while the previous research is only concerned with schooling outcomes, in the current paper we also examine the relationship between ability and child labor.

these effects diminish in magnitude for shocks experienced during the first two years of life, they are still significant and economically meaningful. We find no relationship between rainfall shocks and ability for shocks experienced after age two or before the child was in utero.

The remainder of the paper is organized as follows. Section 2 describes the survey and rainfall data used in the analysis. Section 3 presents our empirical identification strategy and Section 4 presents the main results as well as robustness tests. Section 5 concludes.

2. Nahouri Cash Transfer Pilot Program (NCTPP) Survey and Rainfall Data

2.1 Nahouri Cash Transfer Pilot Program

The Burkina Faso Nahouri Cash Transfer Pilot Program (NCTPP) survey was conducted in June 2008 in Nahouri province in southern Burkina Faso, approximately 100 miles from the capital and bordering Ghana. The survey was the baseline for an ongoing project evaluating social protection strategies in Burkina Faso. Households were randomly selected from a village-level census conducted by our project team immediately prior to the survey in the 75 rural villages of Nahouri that each has a primary school. Households in this region are predominantly rain-fed subsistence farmers growing sorghum and groundnuts and have mean annual per capita expenditures of around \$90.

Our primary analysis focuses on school-aged children 5 to 15 who are biological children of the household head in households with multiple children in this age range and with varying enrollment. There are 2,862 children in 893 different households. As shown in Table 1, 53 percent of this child sample is male and the average age is 9.3 years old. These children live in households with an average of 9.29 individuals, including a head of household, 1.53 wives, 3.97 biological children of the household head ages 5 to 15, 1.25 biological children of the household head under age 5, 0.36 children under age 15 that are not the biological children of the head, and

1.18 other members that include grandparents, aunts, uncles, and other extended family members. Only 12 percent of the sampled children have a parent that ever attended school.

We use the Raven's Colored Progressive Matrices (CPM) to measure a child's cognitive ability. The Raven's CPM is a measure of fluid intelligence or problem solving ability, and it does not require formal schooling to be able to answer the questions (Raven, Raven, and Court 1998). The test does not depend heavily on verbal skills, making it relatively "culture free" (Borghans, Duckworth, Heckman, and Weel 2008). In the Raven's test, the child respondent is asked to select the image that is missing in order to complete a picture. This type of question is novel to the children in Nahouri province, thus providing a truer measure of problem solving skills. We ask 18 questions from the Raven's CPM and on average, children in our sample answer 4.7 questions correctly. Younger children answer fewer questions correctly than older children (the average number correct for children age 5 is 2.7 and for children age 15 is 7.3). To control for this relationship between age and raw test scores, we calculate a z-score for each child measured as the child's raw test score minus the average score for the same age children divided by the standard deviation of test scores for children of that age. Therefore, the mean of the Raven's z-score is zero and the standard deviation is one for each age and across all ages.

In Table 1, we present summary statistics about children's schooling. Few households in rural Burkina Faso ever enroll all of their children. For households with at least two primary school-aged children, 59 percent of these households experience variation in enrollment among their children, 17 percent of these households currently enroll none of their children, and only 24

-

⁶ During extensive pretesting of the Raven's test, results were consistent whether children were asked the entire set of 36 questions or only the odd-numbered questions, so to save interview time we only administered the 18 odd-numbered questions (Sets A, Ab, and B). The average number of questions answered correctly for children ages 6, 7, 8, 9, 10, 11, 12, 13, and 14 is respectively 2.6, 3.6, 4.5, 5.0, 5.5, 5.8, 6.0, 6.4, and 6.3.

⁷ We do not use the international Raven's norming standards since we asked a subset of the Raven's test and what is most important here is how the children in rural Burkina Faso compare to each other, not internationally. Note that in Section 4.3, to test the robustness of the results using the Raven's age-adjusted z-score, we estimate alternative specifications using the Raven's raw test score and results are consistent.

percent of these households currently enroll all of their primary school-aged children. Focusing on the 59 percent of households with variation in enrollment outcomes, which is our baseline sample, 51 percent of children in these households are enrolled in the current school year. If we consider whether a child has ever been enrolled in school rather than current enrollment, then 56 percent of children in these households have ever been enrolled. Given these low enrollment rates, on average these children have only completed 1.7 years of school.

Besides enrollment, we also explore two other school-related outcomes (grade progress and starting school at a late age) where sibling ability rivalry might matter. First, we calculate grade progression through school by dividing the child's highest grade attended by the number of years since the child started school. The grade progress measure ranges from zero to one, with higher numbers indicating quicker progress towards completing primary school. Second, we define a starting school late variable that equals 1 if the child started school after age 7 or never attended school (late start), takes the value 0 if the child started school at age 7 (on-time start), and takes the value -1 or -2 if the child started school at ages 6 or 5, respectively (early start).

In our analysis of child labor activities, we examine both the extensive (whether a child does a specific activity) and intensive margins (how many hours doing an activity). The survey collected information for each child about whether, during the two days preceding the survey when school was in session, the child engaged in different types of child labor activities in the household or on the farm, and if they participated in an activity, during what period of the day they did so. In this rural setting most people do not have watches, and we found during pretesting of the survey instrument that asking about the period during the day for the activity,

⁸ In Section 3.2, we discuss in detail first why we initially restrict the baseline analysis to the sample of households that have within-family variation in enrollment outcomes and second what are the drawbacks of focusing on this selected sample of households that have variation in child schooling. In Section 4.3, we present results incorporating all households and confirm that our main hypotheses about sibling rivalry still hold.

⁹ For children who never attended school, we assign a grade progression measure of zero.

rather than the number of hours, was a more accurate way of understanding how a child's time is allocated. These time periods use the school schedule as a reference period and ask whether a child did a specific activity before school opened, during school hours, during school lunch break, after school closed, all day, or occasionally throughout the day. We use these periods to calculate the number of hours a child engaged in child labor activities. The specific child labor activities we focus on in this paper include: household chores such as cooking, fetching water, sweeping, and doing dishes; tending for siblings or sick members; and engaging in farm labor.

Over 70 percent of children engage in child labor activities and spend on average 5.5 hours per day on them. Sixty percent of children perform household chores, and on average, over 3 hours per day are spent on these chores. There is significant variation in the amount of time spent on chores, suggesting that while the majority of children do them, there are some that are spending an entire day doing so, while other spend a limited time. The average number of hours spent tending for siblings or sick members in the household is 0.6 hours. Twenty-six percent of children engage in this activity, and for those children doing the activity, they spend over 2 hours per day doing so. On average one hour is spent on farm labor and 30 percent of children engage in this activity. For those working on a farm, they do so for an average 3 hours per day.

2.2 Rainfall and ArcGIS Interpolation

To estimate our instrumental variables strategy described in Section 3.2, we need rainfall data for each of the 75 survey villages for each year from 1992 to 2003, which corresponds with the in utero and early childhood years for the primary school-aged children in our survey. Since none of the survey villages have rain station data, but there are rain stations in and around Nahouri

¹⁰ Since the school day starts at 7:30am, ends at 5:00pm, and includes a lunch break of 2.5 hours during which children return home and another 30 minute break at school, we assign the following number of hours to correspond to each time period: 2 (before school opened), 6 (during school), 2 (during school lunch break), 2 (after school closed), 10 (all day), 0.5 (occasionally). To check the robustness of the correspondence between number of hours and time periods during the day, we also estimate several alternative variations, and results are consistent.

province, we use an interpolation procedure that enables us to use these nearby rain stations to estimate rainfall in each village for each year. We purchased rain station data from the Burkina Faso Direction de la Météorologie and from the Ghana Meteorological Agency for the 50 rain stations within 100 kilometers from any of our survey villages (26 in Burkina Faso and 24 in Ghana). On average across the survey villages, the nearest rain station is 9.6 kilometers away.

With the data from the 50 rain stations, we use ArcGIS (geographic information system) to implement an inverse distance weighted (IDW) interpolation procedure, which uses the measured rain data from the stations surrounding a given survey village to predict rainfall in the village. Those rain stations that are closest to the village will have more influence on the predicted value by being weighted more heavily, while those further away will have less influence. 11 The intuition behind the IDW procedure is that locations closer together should have more similar rainfall patterns than those further apart, and therefore in estimating the village's rainfall, additional weight is put on those closer rain stations. ¹² For several reasons, we believe the IDW interpolation procedure provides a more accurate estimated rainfall measure than alternatives such as using the nearest rain station. First, in much of West Africa, there is a strong north-south rain gradient with locations in the north generally receiving less rain than those areas further south. If we were to use only the nearest station and that station were further south than the village of interest, then we would potentially overestimate rainfall in the village. On the other hand, if the closest station were north, then we might underestimate actual rainfall. By using the rain stations in all directions surrounding a given village, this interpolation procedure allows us to account for this rain gradient. Second, if we use the closest rain station to a given village, only

_

any significant topographical features such as mountains that might create areas with a discontinuity in rainfall.

¹¹ Weights are proportional to the inverse distance squared, which is the power that minimizes the root mean squared predicted error and therefore determines the smoothness of the interpolation results. Intuitively, as the distance from the village to the rain station increases, the weight in the interpolation procedure decreases extremely rapidly.

¹² The topography around Nahouri province and the surrounding rain stations consists of gently rolling hills without

four rain stations would be used for 71 of the 75 survey villages. However, there are other rain stations near a village, often in a different direction but slightly further away (for instance, in one village the nearest rain station is 13.45 kilometers and the second closest station is 13.98 kilometers and two other stations are within 18 kilometers). Restricting our analysis to the closest rain station would be discarding valuable information about rainfall patterns around these villages. In contrast, by using information from all rain stations within 100 kilometers of the survey villages and using the IDW interpolation procedure to put more weight on those stations closer to a survey village, we gain a more accurate measure of actual rainfall in each village.

During the years when the sampled children were in utero or early childhood and the 15 years prior to their births, annual rainfall across the villages ranges from 503 to 1388 millimeters, with an mean annual rainfall of 899 millimeters and a standard deviation of 137 millimeters (Table 1). Focusing on the variation within a village over this time period, the standard deviation within a village ranges from 98 to 190 millimeters indicating large year-to-year rainfall fluctuations. To determine if a child was exposed to particularly low rainfall, we compare rainfall in the child's village while in utero or early childhood with that village's historical mean rainfall for the 15 year period (1976-1991) prior to when any of these children were born. Because of our emphasis on the link between low rainfall and poor harvests and the subsequent impact on a child's cognitive ability, we focus on extreme negative shocks and define a negative shock to be rainfall one standard deviation or more below the historical average for a given village.

3. Empirical Identification Strategy

3.1 Previous Sibling Rivalry Research

Sibling rivalry is the idea that within a family, siblings compete for limited resources. If constraints such as credit, capital, or labor bind, all else equal, a child with fewer comparatively

higher valued siblings will be better off. Historically, the sibling rivalry literature for developing countries has focused on sibling sex composition and the number of sisters a child has. This is because in societies with a pro-male bias, investments in girls generally have lower returns than boys, and so having more sisters (i.e. siblings with lower returns) reduces competition for scarce resources and raises investments in all children. Sibling rivalry in child investments in poor countries is well documented (Parish and Willis (1993) for Taiwan; Morduch (2000) for Tanzania; Edmonds (2006) for Nepal; Ota and Moffatt (2007) for India; and Dammert (2010) for Guatemala and Nicaragua).¹³

Although the empirical results from these sibling rivalry studies that focus on sibling sex composition are similar, the underlying behavioral models may differ. For instance, Garg and Morduch (1998) focus on credit constraints and differences in relative investment returns for boys and girls as the cause for sibling rivalry. On the other hand, Edmonds (2006) and Dammert (2010) emphasize that if it is not possible to hire labor for home production and girls have comparative advantage in doing it, both boys and girls benefit from having more sisters (holding constant the number of siblings), an outcome which is observationally indistinguishable from Garg and Morduch (1998). In our paper, by comparing direct measures of child ability across siblings, our approach provides a test of sibling rivalry that does not depend on assumptions of gender bias in returns to education (Garg and Morduch 1998) or on the division of labor based on gender (Edmonds 2007; Dammert 2010).

While these empirical approaches focused on demographic characteristics are useful to highlight tradeoffs parents may face and how they respond to said tradeoffs, these traditional models of sibling rivalry are limited for at least two main reasons. First, they neglect that parents have additional knowledge about their children's capabilities and use this information to make

¹³ Akresh and Edmonds (2011) highlight the distinction between sibling and residential rivalry.

investment decisions. Second, the analysis includes measures of current family size and composition, treating them as given at the time the parents make schooling decisions, despite these variables likely being endogenous. Several recent papers attempt to address these issues by including typically omitted measures of child endowments such as birth weight (Loughran et al. 2004; Rosenzweig and Zhang 2009; Datar et al. 2010), achievement test scores (Glick and Sahn 2010) or cognitive test scores (Ayalew 2005; Kim 2005; Akresh et al. 2012). However, the key challenge in using current measures of child endowments to explain parental investment decisions is that a child's measured endowment (e.g. cognitive ability) is likely endogenous and reflects the entire history of past parental investments, or lack thereof, including nutrition, nurturing, and preferences, as well as any other factors that are unobservable to the researcher.

3.2 Econometric Specification

Our first improvement compared with previous sibling rivalry research is to estimate a household or sibling fixed effects regression that controls for all household level characteristics that are constant across siblings. Specifically, we estimate the following regression:

$$E_{ih} = \beta_0 A_{ih} + \alpha_0 X_{ih} + \lambda_h + \eta_{ih}$$

where E_{ih} is the educational or child labor outcome for child i in household h, A_{ih} is a direct measure of observed child cognitive ability, X_{ih} is a vector of individual characteristics including age and gender that might influence parental investments, λ_h is a household fixed effect that captures all characteristics about the household that are constant across siblings, and η_{ih} is the child specific idiosyncratic error term. In previous sibling rivalry papers, child ability would have been part of the error term, but in our analysis we are able to directly control for its effect on educational and child labor outcomes. This within-family estimate compares a child's own

ability to the average ability of all the other siblings in the household to examine if parents incorporate cognitive ability into the human capital investment decision.¹⁴

While incorporating household fixed effects addresses the issue that family characteristics are different across households, it does not tackle the problem that current child ability measures are endogenous. This endogeneity could be caused by several factors. First, since schooling potentially affects cognitive ability, reverse causality is a concern when using current measures of ability. Second, Raven's test scores not only capture a child's ability endowment but also reflect previous parental investments in the child. In particular, investments in early child nutrition that are critical to cognitive development are reflected to some degree in current ability measures, and it is unlikely these investments are randomly allocated across siblings. Parents with preferences for equity amongst their children may make unequal investments to compensate for initial endowments. Or, parents may prefer to be efficient in their investments, thus investing more in their higher ability children. Any observed measure of a child's endowment, such as ability, therefore could suffer from a potential bias.

Our second empirical identification improvement and what makes this paper unique is that we are able to estimate an instrumental variables model that addresses the endogeneity of child ability. Specifically, we estimate Equation 1 in which we now treat A_{ih} as endogenous, and as instruments, we use rainfall shocks in the child's village for the years when the child was in utero, age zero, and age one. The first stage from this instrumental variables regression is:

-

¹⁴ Much of the previous literature on sibling rivalry uses a household fixed effects conditional logit model to estimate the relationship between enrollment and sibling characteristics, and in that specification all households with no within-family variation in the dependent variable (i.e. households that send all or none of their children to school) will be dropped from the analysis. In order to compare our results with those in the sibling rivalry literature, we initially restrict the analysis to the sample of households with within-family variation in the outcome. While this has the advantage of better linking to the previous literature, we might be overestimating the role of ability in influencing enrollment because by construction the households in this sample already have unequal investments across their children. In Section 4.3, we show that our sibling rivalry results hold even when including households that have no within-family variation in the dependent variable.

(2)
$$A_{ih} = \beta_1 Rainshock_{ih,birthyear-1} + \beta_2 Rainshock_{ih,birthyear} + \beta_3 Rainshock_{ih,birthyear+1} + \alpha_0 X_{ih} + \lambda_h + \varepsilon_{ih}$$

where A_{ih} , X_{ih} , and λ_h are defined as above and $Rainshock_{ih, birthyear-1}$, $Rainshock_{ih, birthyear}$, and $Rainshock_{ih, birthvear+1}$ measure whether child i in household h experienced a negative rain shock in the child's village that could have affected the agricultural harvest for when the child was in utero (birth year minus 1), age zero (birth year), or age one (birth year plus 1). In Section 4.2, we present results showing that these rain shocks are significantly correlated with current measures of a child's cognitive ability. In a setting where families subsist based on agricultural activity, poor rainfall in a given year leads to a bad harvest and subsequent poor nutrition during the year, which will have a direct effect on a child's cognitive development. ¹⁵ The rain shock instruments also likely satisfy the exclusion restriction as they are unlikely to be correlated with the error term in the education or child labor regressions since these rain shocks experienced by a child in utero or under age two occurred 5 to 15 years before the current education or child labor investment decisions. Furthermore, the impact of the rain shocks appears to be captured through the effect on a child's impaired cognitive ability and not overall child health, as we find no correlation between a child's current height-for-age z-score and rain shocks experienced in early childhood.16

¹⁵ All of the instruments focus on annual rainfall shocks for two main reasons. First, we only have year of birth information, but not month of birth, for children in our sample. In rural Burkina Faso, few families obtain official birth certificates because they are expensive, and this was even truer during the period when the children in our sample were born. Second, while the identification strategy theoretically could be sharper by focusing on monthly instead of annual rain shocks (i.e. showing rain shocks one month prior to conception have no impact whereas shocks one month after conception do), the primary mechanism by which rain shocks impact a child's cognitive ability is through nutrition and the agricultural harvest, and what is critical for the size of a harvest is total annual rainfall and not just rainfall in a given month.

¹⁶ A potential threat to our identification strategy is that household crop decisions could render children more or less vulnerable to rainfall shocks. Our instrument would then be correlated with the error term in the second stage regression. We argue, however, that crop decisions depend primarily on landholdings which are time-invariant in rural Burkina Faso because of limited land transactions. Therefore, once we condition on household fixed effects, rainfall shocks should no longer be correlated with the error term in our second stage regression.

4. Empirical Results

4.1 Household Fixed Effects Estimates for Schooling and Child Labor

Since we build on the sibling rivalry literature and to have a baseline with which we can compare our instrumental variables results, we first analyze the within-household relationship between ability, schooling, and child labor outcomes. In Table 2, we present results from estimating the household fixed effects regression in Equation 1. We restrict the household fixed effects specification to the 2,862 school age children (ages 5 to 15) living in households with multiple children in this age range with differing enrollment outcomes. To Consistent with the sibling rivalry literature, we find a strong relationship between a child's cognitive ability and education outcomes. A child with one standard deviation higher ability has a 14.9 percentage point higher likelihood of being currently enrolled compared to their lower ability siblings, corresponding to 29.2 percent of the base enrollment. Similarly, a higher ability child is more likely to ever be enrolled, progress through school more quickly, and is less likely to start school late.

Consistent with the increased schooling results, children with higher ability engage in fewer hours of child labor activities compared to their siblings. Having a standard deviation higher ability is correlated with doing 0.8 fewer hours of child labor than one's lower ability siblings, corresponding to 15.2 percent of average child labor hours worked. This is preliminary evidence that parents consider a child's cognitive ability in deciding how to allocate their children's time. Ability and child labor are strongly correlated when considering the intensive margin of number of hours performing specific child labor activities (household chores and tending for siblings or sick members), but there is no relationship when considering the extensive margin of whether a child engages in specific child labor tasks. This suggests that both high and

-

¹⁷ All regressions include household fixed effects, as well as child age and gender dummies. Correlation among the error terms for children in a given village experiencing the same enrollment and rainfall shock environment might bias the standard errors downward, so in all regressions we cluster the standard errors by village.

low ability children engage in these different tasks, and that ability may not be protective of a child's time in this regard, rather it mainly affects the number of hours a child performs these tasks. However, as discussed earlier, these estimates may be biased because measured ability for school-age children will reflect the sum of all previous investments and shocks for that child.

4.2 Household Fixed Effects IV Estimates for Schooling and Child Labor

Since measured ability reflects the accumulation of previous parental investments in a particular child, the OLS fixed effect estimates in Table 2 might have an upward bias in estimating the relationship between ability and enrollment or child labor. On the other hand, the OLS estimates may be biased downwards because of negative shocks experienced during early childhood.

Using an instrumental variables (IV) approach that treats child ability as endogenous provides a causal estimate of the effect of a child's ability on enrollment and child labor. Our IV approach takes advantage of the relationship between rainfall, food scarcity, and child cognitive development. As discussed earlier, malnourishment has large effects on healthy development, particularly for young children. The economics and nutrition literatures suggest that, while all early years are important, the periods before a child is two and in utero are particularly important to development. We therefore explore various rain shock instruments from this early childhood time period, with a particular focus on the child's years in utero, age zero, and age one.

Table 3 presents results from the IV regression's first stage of the endogenous variable, child cognitive ability, on alternative rain shock instruments. The instruments measure if rainfall in a given year in a child's village is at least one standard deviation below that village's historical mean rainfall for the 15-year period before any of the children in the survey were born. Negative rainfall reduces a harvest and has an adverse effect on a child's food consumption or nutrition. In Columns 1 to 7, we examine the relationship between ability and each shock time period

separately; in Column 8, we combine the instruments for rain shocks experienced while a child was in utero, age 0, or age 1 as in Equation 2. A large negative rain shock for the harvest consumed by the mother two or three years before birth has no relationship with a child's cognitive ability. Obviously, this is before the child's conception, and while the mother could be affected by the shock, we find no correlation with the child's ability.

However, having a large negative shock for the harvest consumed by the mother during the child's in utero year has a large negative correlation with the child's cognitive development as seen in column 3. This is consistent with the previous literature regarding the importance of a mother's health during pregnancy. Having rainfall in the child's village for the harvest when the child is in utero that is more than one standard deviation below the village's mean historical level is correlated with a 0.23 standard deviations lower cognitive ability z-score. The F-statistic for the excluded instrument is well above the threshold that would indicate a potential weak instrument bias. We observe a similar relationship in column 4 for a rain shock for the harvest during the child's first year of life (age 0). Experiencing a rain shock when the child is age 1 is negatively correlated with ability, with a similar size impact as seen for shocks of children age 0 and in utero, but the estimate is not significant at standard levels. Columns 6 and 7 present first stage regressions for the harvests for children ages 2 to 5, and the point estimates are small and insignificant, suggesting a negative shock during these ages is less important than earlier years.

Since the child's in utero, age 0, and age 1 periods are shown to be important (in the literature and in our first stage regressions), we estimate the first stage using those three instruments in Column 8. Point estimates are large and significant, with an F-statistic for the excluded instruments of 9.69, suggesting this strategy is valid. Rainfall for a child in utero that is more than one standard deviation below the village's long-run historical mean is correlated with

0.24 standard deviations lower ability. A rain shock for a child age 0 is correlated with 0.15 standard deviations lower ability, while a rain shock experienced by a child age 1 is correlated with 0.27 standard deviations lower ability.

In Table 4, we present results from the household fixed effects instrumental variables regressions estimating the relationship between child cognitive ability and education outcomes and treating ability as endogenous. Results are consistent with the Table 2 OLS estimates, although the magnitude of the effects are larger, which we discuss in detail below. In treating child ability as endogenous, we use the three rain shock instruments for the time periods in utero, age zero, and age one. A one standard deviation increase in child ability is correlated with an 81.7 percentage point higher likelihood of enrollment. To put this in perspective, a child that experiences a negative rain shock (rainfall one standard deviation below the village's historical long-run average) while in utero would have 0.235 standard deviations lower cognitive ability and this would lead to a 19.2 percentage point lower likelihood of enrollment, corresponding to a 38 percent drop relative to mean enrollment. Similar calculations show that negative rain shocks experienced by a child at age 0 correspond to a 23 percent drop in enrollment, while rain shocks experienced by a child at age 1 correspond to a 44 percent enrollment drop.

The Table 4 IV estimates are substantially larger in magnitude than the Table 2 OLS estimates. In addition to eliminating the endogeneity bias present in the OLS estimate, one potential explanation for this difference is that, if there is heterogeneity in the effects of ability, the IV estimates obtained are not consistent estimates of the average effect of ability on

-

¹⁸ Since the model is overidentified, we are able to test whether the instruments are uncorrelated with the error term and correctly excluded from the estimated enrollment equation. The overidentification test regresses the residuals from the IV regression on all of the instruments, and under the null hypothesis that all instruments are uncorrelated with the error term, a rejection of Hansen's J statistic casts doubt on the validity of the instruments. For the enrollment regression, the p-value of the J statistic is 0.3 indicating our instruments are valid, while for the other dependent variables in Tables 4 and 5, the p-values of the J statistic are generally slightly larger.

¹⁹ Results are consistent when we include in the regressions additional child level controls such as birth order or a child's height-for-age z-score.

enrollment in the overall population of children (Card 1995, 2001; Heckman et al. 1999). Rather, the IV estimate should be interpreted as the local average treatment effect, meaning the effect of ability on enrollment for the subgroup of children from households for which rainfall shocks to a child in utero or during early childhood caused ability to differ across siblings (Imbens and Angrist 1994). Children whose ability has been negatively affected by these early childhood shocks are likely to be from vulnerable families who did not have safety nets to buffer the shocks. Presumably, those are also families where child ability, relative to his siblings' ability, plays a crucial role in deciding school enrollment.²⁰ For example, while in a less vulnerable family most children would be enrolled, in more vulnerable families, only the most able children would get a chance to enroll. This could explain why the local average treatment effects obtained from the IV regressions are larger than the OLS estimates, which are estimates for the overall population of children.

The other Table 4 columns present the household fixed effects IV regressions for ever being enrolled, timely grade progression through school, and starting school at a late age. A negative shock while in utero that lowers child cognitive ability by 0.235 standard deviations would lead to declines, relative to baseline levels, of 32 percent for ever being enrolled, 24 percent for timely grade progression, and a 113 percent increase in starting school late.²¹

Turning to child labor activities in Table 5, for the same sample of 2,862 children ages 5 to 15 years old, higher ability children engage in less child labor than their lower ability siblings.

Negative rain shocks experienced while in utero lowers cognitive ability by 0.235 standard

When we split households into poor and non-poor groups based on assets, per capita expenditures, or parental schooling, the relationship between child ability and enrollment in Table 4 only holds for poor households.
To further explore the relationship between the specific timing of rainfall shocks and education, in Appendix Table 1, we present the household fixed effects IV results using each instrument (rain shock experienced while in utero, age 0, or age 1) separately. Results for each instrument are consistent with those in Table 4, indicating that higher ability siblings within a household, compared to their lower ability siblings, are more likely to receive positive schooling investments. The magnitude of the impact of ability on enrollment is largest for those children who experience negative rainfall shocks while in utero and diminishes for those experiencing shocks at later ages.

deviations and leads to increases for these children of 6.96 percentage points in the likelihood of tending for siblings or sick members and 6.56 percentage points in the likelihood of working on the family farm. Relative to baseline levels, these shocks represent increases of 26.8 and 21.9 percent, respectively compared to a child's higher ability siblings. Even though a child is not more likely to perform chores around the household such as cooking, fetching water, sweeping or doing the dishes, a child with 0.235 standard deviations lower cognitive ability due to negative rain shocks in utero will spend 0.65 more hours on said tasks than his higher ability siblings, which is 21.2 percent of the baseline level. Similar calculations show that a negative rain shock in utero leads to declines in ability and these further lead to increases of 0.24 hours per day tending for siblings or sick household members, 1.19 hours per day on farm labor, and 2.71 total hours of child labor per day. Relative to baseline mean levels, these additional hours represent increases of 39.3, 126.6, and 48.9, respectively. Experiencing a 1 standard deviation rainfall shock at age 0 or age 1 corresponds respectively to 0.146 or 0.272 standard deviations lower cognitive ability and subsequently 30.4 or 56.6 percent increases in the number of hours of child labor these children perform relative to their higher ability siblings.²²

4.3 Robustness Checks

To test the robustness of our results, we present alternative specifications in which we use placebo instruments and a different measure of child ability. We also note that households that had children who experienced negative rain shocks are not systematically different in observable characteristics from other households and that restricting the sample based on additional

-

²² Similar to Appendix Table 1, in Appendix Table 2, we explore the relationship between the specific timing of rainfall shocks and child labor outcomes using each rain shock instrument separately. Overall, results are generally consistent with those in Table 5, indicating that higher ability children perform less hours of child labor, although the results for the extensive margin of whether a child engages in farm labor are less consistent and we observe smaller magnitude impacts for the intensive margin of child labor hours for shocks experienced in utero compared to the education regressions. This is consistent with in utero shocks impacting brain development of the fetus and subsequent cognitive ability and therefore having a larger impact on schooling-related outcomes.

information about a child's birth village yields consistent results. Finally, we discuss results from regressions that include the sample of households with no within-family variation in the dependent variable, as well as regressions estimated without household fixed effects, both of which support our hypotheses about sibling rivalry.

In Table 6, we provide suggestive evidence that our rain shock instruments likely satisfy the exclusion restriction. To show this, we use a placebo instrument that uses a rain shock for the harvest two years before birth (a period that has no link to child ability as seen in Table 3 column 2) to rule out the possibility that rain shocks have long-lasting and direct effects on education and labor outcomes. The F-statistic for the excluded instrument is 0.97 and the coefficient for the instrument is insignificant, meaning it is uncorrelated with child ability. We present results for the second stage, looking at the same schooling and child labor outcomes, and we observe no statistically significant relationship between ability and these outcomes.²³ This strengthens our argument that our results in previous tables using valid instruments are indeed estimating a causal relationship.

To address concerns that transforming the Raven's test scores into age-adjusted z-scores might have introduced bias, we estimate regressions using the Raven's raw test score in Table 7. A negative rain shock experienced in utero lowers a child's Raven's raw test score by 0.716 questions (in the first stage results that are not shown) and this would lead to an 18.9 percentage point lower likelihood of current enrollment and 2.53 hours more child labor, which are consistent with the results presented earlier in Tables 4 and 5.

Another concern is there could be selection bias in the types of households that have a child during periods of negative shocks, and these households could be systematically different

²³ In addition, shocks for the harvest three years prior to birth have no link to ability as shown in Table 3 column 1, and the second stage estimates are similar to those in Table 6.

from households that did not have children during those shock periods. In Table 8, we present regressions of different observable household characteristics measured at the time of the survey on an indicator for whether the household had a child who experienced a negative rainfall shock in utero or during early childhood.²⁴ We examine per capita household expenditures, log assets, parental schooling, household size, and the household head's marital status, age, occupation, number of wives, and number of children.²⁵ There is no evidence that households that had a child during periods of negative rain shocks are any different, at least on observable dimensions, from other households, as the coefficient for the shock indicator is insignificant in each specification.

In creating our rainfall shock instruments, we have assumed that all children were born in the same village in which they currently reside. However, due to previous migration, if a child was born in a different village, then the instrument that is being used for that child's in utero and early childhood rainfall is measured with error. We therefore test the robustness of our results and correct for this potential problem using the limited information we have about each person's birth location. Using two alternative specifications, we limit the sample to exclude those children for whom we have varying levels of information about whether they were born in the current village. In the first specification, we use all information we have to identify anyone we believe was not born in the same village in which they currently reside, and we then drop them from the sample, reducing our sample size by 1.3 percent. In this approach, we only drop those for whom we are confident they were born in a different village. In a second, more restrictive, specification, we drop anyone for whom we do not have sufficient information to determine they

²⁴ If we had complete birth history information for each household, we could estimate these regressions using an indicator for whether the household gave birth to a child during a negative rain shock period, as this would allow us to accurately capture potential endogenous fertility. Given we do not have this information, we are therefore limited in measuring households that gave birth during a shock period and had a child survive until the survey.

²⁵ While we would like to estimate these regressions using characteristics measured at the time of the rain shock, we do not have that information in the survey. However, some of the variables are unlikely to have changed since the time period of the shock, including parental schooling and the household head's occupation.

were born in the village where they are currently residing, thus limiting our sample by 5.4 percent. Results for both specifications (not shown) are consistent with those presented thus far.

Thus far, to better link our results with the previous literature, all regressions exclude the sample of households with no within-family variation in the dependent variable. However, by doing that the remaining sample of households is a selective one and by focusing on households that already have unequal investments across their children, we might be overestimating the role of ability in influencing enrollment and child labor. As a robustness check, we estimate the household fixed effects IV regressions on the full sample of households, including those who have enrolled all or none of their children. The results (not shown) indicate a consistent pattern of sibling ability rivalry, whereby higher ability children are more likely to be enrolled and work fewer hours doing child labor compared to their lower ability siblings. Finally, all regressions estimated so far incorporate household fixed effects, which have the significant advantage of controlling for all observable and unobservable household characteristics that are constant across siblings and that might be related to education and child labor outcomes. As a robustness check, we re-estimate the IV regressions for current enrollment and total child labor hours and replace the household fixed effects with village fixed effects. Identification is no longer driven by within household variation across siblings but also includes across household variation. Results (not shown) illustrate that higher ability children are more likely to be enrolled and spend fewer hours doing child labor compared to lower ability children, supporting our previous results about the relationship between ability and human capital investments.

5. Conclusions

In this paper, we find strong evidence of sibling rivalry when parents make decisions regarding educational investments and child labor in rural Burkina Faso. However, in contrast with

previous research that generally focuses on easily observable demographic characteristics to gauge sibling rivalry, we use measures of a child's own cognitive ability to test for how parents make these investment decisions. To address the potential endogeneity of current measures of child ability, we use rain shocks in a child's village that were experienced in utero or during early childhood to instrument for cognitive ability. We examine both educational outcomes (enrollment, grade progression, and late start) as well as child labor, focusing not only on total hours of labor, but also changes in the specific tasks that each child performs. Negative rainfall shocks experienced in utero lead to 0.24 standard deviations lower ability z-scores and compared to their siblings that corresponds with a 38 percent drop in enrollment and a 49 percent increase in child labor hours. Our findings are robust to alternative education and child labor outcomes, as well as alternative child ability measures. Negative education impacts are largest for shocks experienced in utero, diminished for shocks before age two, and have no impact for shocks after age two.

Our results can likely be generalized to other developing countries that have not yet achieved universal primary or secondary education and where families face real resource constraints that affect investment decisions related to education and child labor. Child labor activities such as farm labor, tending for siblings, and chores around the house are common throughout poor countries, and children are often required to work long hours doing these activities.

The policy implications of our results are twofold. First, our results show that, when faced with severe resource constraints, parents act strategically when investing in their children's human capital, making choices based on the ability of their children. However, those are hard choices, which are leaving behind many children who could still benefit from attending school.

Using our estimates of the impact of sibling ability rivalry on educational investments, we can speculate on the long-term consequences that follow from these hard choices. Lower ability children are 32 percent less likely ever to attend school compared to their higher ability siblings, and this would translate, in this setting, into missing on average 3.8 years of school by the time they were age 16. Using a 9.9 percent rate of return for each year of primary school in Burkina Faso (Kazianga 2004), this corresponds to a 12 percent loss in future earnings. Programs such as cash transfers that would relax the resource constraints for those households might be useful to increase enrollment.

The second policy implication is that negative shocks early in a child's life, particularly in utero or in the first years of life, can have long-run implications for that child's wellbeing. If a child is exposed to a negative rainfall shock in utero in an environment where families are subsisting on their rain-fed agricultural production, the likelihood that the child will ever recover from said shock is low. Policies to prevent this type of food and nutrition insecurity are critical. Likewise, policies aimed at improving child development may improve long-run well-being through the direct effect on child development at the time and the subsequent effect on parent investments, which could reduce the inefficiencies generated by sibling ability rivalry.

References

- Aizer, Anna. 2004. "Home Alone: Supervision After School and Child Behavior." *Journal of Public Economics*, 88(9): 1835-1848.
- Akresh, Richard and Eric Edmonds. 2011. "Residential Rivalry and Constraints on the Availability of Child Labor." NBER Working Paper 17165.
- Akresh, Richard, Emilie Bagby, Damien de Walque, and Harounan Kazianga. 2012. "Child Ability and Household Human Capital Investment Decisions in Burkina Faso." *Economic Development and Cultural Change*, 61(1): 157-186.
- Akresh, Richard, Leonardo Lucchetti, and
- Alderman, Harold, John Hoddinott, and Bill Kinsey. 2006. "Long Term Consequences of Early Childhood Malnutrition." *Oxford Economic Papers*, 58(3): 450-474.
- Almond, Douglas and Janet Currie. 2011. "Killing Me Softly: The Fetal Origins Hypothesis." *Journal of Economic Perspectives*, forthcoming.
- Almond, Douglas. 2006. "Is the 1918 Influenza Pandemic Over? Long-Term Effects of In Utero Influenza Exposure in the Post-1940 U.S. Population." *Journal of Political Economy*, 114(4): 672-712.
- Ayalew, Tekabe. 2005. "Parental Preference, Heterogeneity, and Human Capital Inequality." *Economic Development and Cultural Change*, 53(2): 381-407.
- Barham, Tania. 2012. "Enhancing Cognitive Functioning: Medium-Term Effects of a Health and Family Planning Program in Matlab." *American Economic Journal: Applied Economics*, 4(1): 245-273.
- Barker, D.J.P. 1998. *Mothers, Babies, and Health in Later Life*. Edinburgh, U.K.: Churchill Livingstone.
- Basu, Kaushik. 1999. "Child Labor: Cause, Consequence, and Cure, with Remarks on International Labor Standards." *Journal of Economic Literature*, 37(3): 1083-1119.
- Becker, Gary S. and Nigel Tomes. 1976. "Child Endowments and the Quantity and Quality of Children." *Journal of Political Economy*, 84(2): S143-S162.
- Beegle, Kathleen, Rajeev Dehejia, and Roberta Gatti. 2009. "Why Should We Care About Child Labor? The Education, Labor Market, and Health Consequences of Child Labor." *Journal of Human Resources*, 44(4): 871-889.
- Behrman, Jere R. and Mark R. Rosenzweig. 2004. "Returns to Birthweight." *Review of Economics and Statistics*, 86(2): 586-601.

- Black, Sandra, Paul Devereux, and Kjell Salvanes. 2005. "The More the Merrier? The Effect of Family Size and Birth Order on Children's Education." *Quarterly Journal of Economics*, 120(2): 669-700.
- Boozer, Michael and Tavneet Suri. 2001. "Child Labor and Schooling Decisions in Ghana." Yale University, manuscript.
- Borghans, Lex, Angela Lee Duckworth, James A. Heckman, and Bas ter Weel. 2008. "The Economics and Psychology of Personality Traits." NBER Working Paper 13810.
- Bundervoet, Tom, Philip Verwimp, and Richard Akresh. 2009. "Health and Civil War in Rural Burundi." *Journal of Human Resources*, 44(2): 536-563.
- Card, David. 1995. "Earnings, Schooling, and Ability Revisited." In *Research in Labor Economics*, volume 14, Editor Solomon Polachek, Greenwich Connecticut: JAI Press, 23-48.
- Card, David. 2001. "Estimating the Return to Schooling: Progress on Some Persistent Econometric Problems." *Econometrica*, 69(5): 1127-1160.
- Dammert, Ana. 2010. "Siblings, Child Labor, and Schooling in Nicaragua and Guatemala." *Journal of Population Economics* 23(1): 199-224.
- Datar, Ashlesha, M. Rebecca Kilburn, and David S. Loughran. 2010. "Endowments and Parental Investments in Infancy and Early Childhood." *Demography*, 47(1): 145-162.
- Edmonds, Eric and Nina Pavcnik. 2005. "Child Labor in the Global Economy." *Journal of Economic Perspectives*, 19(1): 199-220.
- Edmonds, Eric. 2006. "Understanding Sibling Differences in Child Labor." *Journal of Population Economics*, 19(4): 795-821.
- Edmonds, Eric. 2008. "Child Labor." In T. P. Schultz and J. Strauss, *eds.*, *Handbook of Development Economics Volume 4*, (Elsevier Science, Amsterdam, North-Holland).
- Emerson, Patrick M. and Andre Portela Souza. 2008. "Birth Order, Child Labor, and School Attendance in Brazil." *World Development*, 36(9): 1647-1664.
- Field, Erica, Omar Robles, and Maximo Torero. 2009. "Iodine Deficiency and Schooling Attainment in Tanzania." *American Economic Journal: Applied Economics*, 1(4): 140-169.
- Garg, Ashish and Jonathan Morduch. 1998. "Sibling Rivalry and the Gender Gap: Evidence from Child Health Outcomes in Ghana." *Journal of Population Economics*, 11(4): 471-493.

- Glewwe, Paul and Michael Kremer. 2006. "Schools, Teachers, and Education Outcomes in Developing Countries." In *Handbook on the Economics of Education*, editors Erik Hanushek and Finis Welch. Amsterdam: North-Holland.
- Glick, Peter and David E. Sahn. 2010. "Early Academic Performance, Grade Repetition, and School Attainment in Senegal: A Panel Data Analysis." *World Bank Economic Review*, 24(1): 93-120.
- Grantham-McGregor, Sally, Lia Fernald, and Kavita Sethuraman. 1999. "Effects of Health and Nutrition on Cognitive and Behavioral Development in Children in the First Three Years of Life. Part 1: Low Birthweight, Breastfeeding, and Protein-energy Malnutrition." *Food and Nutrition Bulletin*, 20(1): 53-75.
- Heckman, James, Robert Lalonde, Jeffrey Smith. 1999. "The Economics and Econometrics of Active Labor Market Programs." In Orley Ashenfelter and David Card, editors, *Handbook of Labor Economics*, volume 3, Elsevier Science, Amsterdam.
- Imbens, Guido and Joshua Angrist. 1994. "Identification and Estimation of Local Average Treatment Effects." *Econometrica*, 62(2): 467-475.
- Kazianga, Harounan. 2004. "Schooling Returns for Wage Earners in Burkina Faso: Evidence from the 1994 and 1998 National Surveys." Yale University Economic Growth Center Discussion Paper 892.
- Kessler, Daniel. 1991. "Birth Order, Family Size, and Achievement: Family Structure and Wage Determination." *Journal of Labor Economics*, 9(4): 413-426.
- Kim, Hisam. 2005. "Parental Investment Between Children With Different Abilities." University of Wisconsin, Madison, manuscript.
- Knudsen, E. I. 2004. "Sensitive Periods in the Development of the Brain and Behavior." *Journal of Cognitive Neuroscience*, 16(8): 1412-1425.
- Levison, Deborah, Karine Moe, and Felicia Marie Knaul. 2001. "Youth Education and Work in Mexico." *World Development*, 29(1): 167-188.
- Loughran, David S., Ashlesha Datar, and M. Rebecca Kilburn. 2004. "The Interactive Effect of Birth Weight and Parental Investment on Child Test Scores." RAND Working Paper 168.
- Maccini, Sharon and Dean Yang. 2009. "Under the Weather: Health, Schooling, and Economic Consequences of Early-Life Rainfall." *American Economic Review*, 99(3): 1006-1026.
- Macours, Karen, Norbert Schady, and Renos Vakis. 2012. "Cash Transfers, Behavioral Changes, and Cognitive Development in Early Childhood: Evidence from a Randomized Experiment." *American Economic Journal: Applied Economics*, 4(2): 247-273.

- Maluccio, John, John Hoddinott, Jere Behrman, Reynaldo Martorell, Agnes Quisumbing, and Aryeh Stein. 2009. "The Impact of Improving Nutrition During Early Childhood on Education Among Guatemalan Adults." *Economic Journal*, 119(537): 734-763.
- Morduch, Jonathan. 2000. "Sibling Rivalry in Africa." *American Economic Association Papers and Proceedings*, 90(2): 405-409.
- Oreopoulos, Philip, Mark Stabile, Randy Walld, and Leslie Roos. 2008. "Short-, Medium-, and Long-Term Consequences of Poor Infant Health: An Analysis Using Siblings and Twins." *Journal of Human Resources*, 43(1): 88-138.
- Ota, Masako, and Peter Moffatt. 2007. "The Within-household Schooling Decision: A Study of Children in Rural Andhra Pradesh." *Journal of Population Economics* 20(1): 223-239.
- Ozier, Owen. 2011. "Exploiting Externalities to Estimate the Long-Term Effects of Early Childhood Deworming." University of California, Berkeley, manuscript.
- Parish, William and Robert Willis. 1993. "Daughters, Education, and Family Budgets: Taiwan Experiences." *Journal of Human Resources*, 28(4): 863-898.
- Raven, John E., John C. Raven, and John H. Court. 1998. *Manual for Raven's Progressive Matrices and Vocabulary Scales: Section 1 General Overview*. Oxford: Oxford Psychologists Press.
- Rosenzweig, Mark R. and Junsen Zhang. 2009. "Do Population Control Policies Induce More Human Capital Investment? Twins, Birth Weight and China's 'One-Child' Policy." *Review of Economic Studies*, 76(3): 1149-1174.
- Royer, Heather. 2009. "Separated at Girth: Estimating the Long-Run and Intergenerational Effects of Birthweight Using Twins." *American Economic Journal: Applied Economics*, 1(1): 49–85.
- Stein, Zena, Mervyn Susser, Gerhart Saenger, and Francis Marolla. 1975. *Famine and Human Development: The Dutch Hunger Winter of 1944-1945*. New York: Oxford University Press.
- Strauss, John and Duncan Thomas. 1995. "Human Resources: Empirical Modeling of Household and Family Decisions." In *Handbook of Development Economics Volume 3A*, editors Jere Behrman and T.N. Srinivasan. Amsterdam: Elsevier/North Holland.
- Strauss, John and Duncan Thomas. 2008. "Health Over the Life Course." In *Handbook of Development Economics*, vol. 4, editors Paul Schultz and John Strauss, Amsterdam: North-Holland, 3375-3474.
- United Nations. 2010. "The Millennium Development Goals Report 2010" United Nations, New York.

Table 1: Summary Statistics of Nahouri Cash Transfer Pilot Program (NCTPP) Survey

| | Mean | Standard |
|---|-------|-----------|
| | | Deviation |
| Household Size | 9.29 | 3.91 |
| Number of Wives | 1.53 | 0.95 |
| Number of Biological Children Ages 5-15 | 3.97 | 1.75 |
| Number of Biological Children Under Age 5 | 1.25 | 1.14 |
| Number of Non-Biological Children in Household | 0.36 | 0.85 |
| Number of Other Members (Excluding Head) | 1.18 | 1.46 |
| Male (Fraction Male) | 0.53 | 0.50 |
| Age | 9.26 | 3.03 |
| Own Ability (Raven's age adjusted z-score) | -0.02 | 1.00 |
| Raven's Raw Test Score | 4.74 | 3.36 |
| Proportion Children Currently Enrolled | 0.51 | 0.50 |
| Proportion Children Ever Enrolled | 0.56 | 0.50 |
| Grade Progression | 0.49 | 0.48 |
| Starting School Late | 0.34 | 0.97 |
| Binary Indicator if Child Does Household Chores such as | 0.60 | 0.49 |
| Cooking, Fetching Water, Sweeping, Doing Dishes | | |
| Binary Indicator if Child Tends for Siblings or Sick | 0.26 | 0.44 |
| Members in the Household | | |
| Binary Indicator if Child Engages in Farm Labor | 0.30 | 0.46 |
| Hours Doing Household Chores such as Cooking, Fetching | 3.06 | 5.51 |
| Water, Sweeping, Doing Dishes | | |
| Hours Tending for Siblings or Sick Members | 0.61 | 2.02 |
| Hours Engaging in Farm Labor | 0.94 | 2.41 |
| Total Hours Engaging in Child Labor Activities | 5.54 | 9.60 |
| Number of Children | 2862 | |
| Number of Households | 893 | |
| Annual Village Rainfall from 1976-2003 (mm) | 899 | 137 |

Notes: All summary statistics are based on information for the 2862 children ages 5-15 from the 893 households that have multiple children in this age range with differing enrollment outcomes. Own ability is measured using the Raven's Colored Progressive Matrices and normed by age (z-score); grade progression in school is the child's grade in school divided by number of years since the child started attending school and ranges from 0 to 1; starting school late equals 1 if a child started school after age 7 or never attended school, 0 if started school at age 7, -1 if started school at age 6, and -2 if started school at age 5. Each of the binary indicators for child labor activity (does household chores, tends for siblings or sick members, engages in farm labor) equals 1 if the child engaged in that activity during the 2 days preceding the survey when school was in session. Rainfall summary statistics use annual rainfall data for the 75 villages in the NSPP survey. From 1976-2003, across all 75 villages, annual minimum rainfall was 503mm and maximum rainfall was 1388mm. Data source: Nahouri Cash Transfer Pilot Program (NCTPP) 2008 household survey data and rainfall data from Burkina Faso and Ghana Meteorological Services.

| Table 2: Household Fixed Effects Regressions Estimating the Relationship Between Ability, Schooling, and Child Labor | | | | | | | | | |
|--|------------|---------------|-------------|-------------|-----------------|----------|-------------|--|--|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | | |
| Panel A. Education | | | | | | | | | |
| Dependent Variable: | Current | Ever Enrolled | Grade | Late Start | | | | | |
| | Enrollment | | Progression | | | | | | |
| Ability (Raven's age | 0.149*** | 0.139*** | 0.142*** | -0.273*** | | | | | |
| adjusted z-score) | [0.015] | [0.013] | [0.015] | [0.030] | | | | | |
| Male | 0.044** | 0.047** | 0.057*** | -0.069 | | | | | |
| | [0.019] | [0.019] | [0.020] | [0.046] | | | | | |
| Household Fixed Effects? | Yes | Yes | Yes | Yes | | | | | |
| Age Fixed Effects? | Yes | Yes | Yes | Yes | | | | | |
| Observations | 2,862 | 2,862 | 2,714 | 2,862 | | | | | |
| Panel B. Child Labor | | | | | | | | | |
| Dependent Variable: | Does | Tends for | Engages in | Hours Doing | Hours Tending | Hours in | Total Child | | |
| | Household | Siblings or | Farm | Household | for Siblings or | Farm | Labor Hours | | |
| | Chores | Sick Members | Labor | Chores | Sick Members | Labor | | | |
| Ability (Raven's age | -0.006 | -0.006 | -0.012 | -0.455*** | -0.141** | -0.095 | -0.840*** | | |
| adjusted z-score) | [0.011] | [0.008] | [0.012] | [0.143] | [0.063] | [0.068] | [0.244] | | |
| Male | -0.249*** | -0.129*** | 0.109*** | -3.039*** | -0.418*** | 0.435*** | -3.079*** | | |
| | [0.024] | [0.019] | [0.014] | [0.260] | [0.094] | [0.096] | [0.374] | | |
| Household Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes | | |
| Age Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes | | |
| Observations | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | | |

Notes: Robust standard errors in brackets, clustered at village level. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include household fixed effects and child age and gender dummies. Regression sample includes 2862 children ages 5-15 from households with multiple siblings in this age range with differing enrollment outcomes. Ability is measured using the Raven's Colored Progressive Matrices and normed by age (z-score). All dependent variables are defined in the notes to Table 1. Data source: Nahouri Cash Transfer Pilot Program (NCTPP) 2008 household survey data.

Table 3: First Stage Household Fixed Effects IV Regressions: Relationship between Potential Rain Shock Instruments and Child Ability

| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) |
|----------------------------|---------|---------|-----------|-----------|---------|---------|---------|-----------|
| Dependent Variable: | Ability | Ability | Ability | Ability | Ability | Ability | Ability | Ability |
| Rain Shock for Harvest For | 0.048 | | | | | | | |
| Age 3 Years Before Birth | [0.145] | | | | | | | |
| Rain Shock for Harvest For | | 0.191 | | | | | | |
| Age 2 Years Before Birth | | [0.194] | | | | | | |
| Rain Shock for Harvest For | | | -0.228*** | | | | | -0.235*** |
| In Utero | | | [0.047] | | | | | [0.058] |
| Rain Shock for Harvest For | | | | -0.208*** | | | | -0.146* |
| Age 0 | | | | [0.054] | | | | [0.084] |
| Rain Shock for Harvest For | | | | | -0.248 | | | -0.272* |
| Age 1 | | | | | [0.171] | | | [0.158] |
| Rain Shock for Harvest For | | | | | | -0.067 | | |
| Ages 2 or 3 | | | | | | [0.070] | | |
| Rain Shock for Harvest For | | | | | | | 0.022 | |
| Ages 4 or 5 | | | | | | | [0.216] | |
| Household Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Gender Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Age Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| F-statistic of Excluded | 0.11 | 0.97 | 23.95 | 15.03 | 2.11 | 0.94 | 0.01 | 9.69 |
| Instruments | | | | | | | | |
| Observations | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 |

Notes: Robust standard errors in brackets, clustered at village level. * significant at 10%; ** significant at 5%; *** significant at 1%. Regression results are from the instrumental variables first stage regression of ability on alternative rain shocks. All regressions include household fixed effects and child age and gender dummies. Regression sample includes 2862 children ages 5-15 from households with multiple siblings in this age range with differing enrollment outcomes. Ability is measured using the Raven's Colored Progressive Matrices and normed by age (z-score). Instruments for each column vary by the year relative to the child's birth year, and consider if rainfall for the harvest for that year in the child's village is 1 standard deviation below the village's historical average rainfall for the 15-year period before any of the survey children were born (1976-1991). Data source: Nahouri Cash Transfer Pilot Program (NCTPP) 2008 household survey data and rainfall data from Burkina Faso and Ghana Meteorological Services.

Table 4: Household Fixed Effects IV Regressions Estimating Relationship between Ability and Schooling,
Treating Ability as Endogenous

| | (1) | (2) | (3) | (4) |
|---|--|--|--|--|
| Dependent Variable: | Current Enrollment | Ever Enrolled | Grade Progression | Late Start |
| Instruments: | Rain Shocks for Harvests for In Utero, Age 0, or Age 1 | Rain Shocks for Harvests for In Utero, Age 0, or Age 1 | Rain Shocks for Harvests for In Utero, Age 0, or Age 1 | Rain Shocks for Harvests for In Utero, Age 0, or Age 1 |
| Ability (Raven's age adjusted z-score) | 0.817*** [0.273] | 0.771*** [0.175] | 0.501*** [0.159] | -1.637* [0.863] |
| Male | -0.029 [0.045] | -0.022 [0.036] | 0.017 [0.033] | 0.080 [0.117] |
| Household Fixed Effects? | Yes | Yes | Yes | Yes |
| Age Fixed Effects? | Yes | Yes | Yes | Yes |
| First Stage F-Statistic of Excluded Instruments | 9.69 | 9.69 | 9.64 | 9.69 |
| Observations | 2,862 | 2,862 | 2,714 | 2,862 |

Notes: Robust standard errors in brackets, clustered at village level. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include household fixed effects and child age and gender dummies. Regression sample includes 2862 children ages 5-15 from households with multiple siblings in this age range with differing enrollment outcomes. Ability is measured using the Raven's Colored Progressive Matrices and normed by age (z-score). All dependent variables are defined in the notes to Table 1. The 3 instruments measure if rainfall for the harvest for the in utero, age 0, or age 1 years in the child's village are 1 standard deviation below the historical average rainfall for the 15-year period before any of the survey children were born (1976-1991). Data source: Nahouri Cash Transfer Pilot Program (NCTPP) 2008 household survey data and rainfall data from Burkina Faso and Ghana Meteorological Services.

Table 5: Household Fixed Effects IV Regressions Estimating Relationship between Ability and Child Labor,
Treating Ability as Endogenous

| | (1) | (2) | $1119 \times 1111 \times 11111 \times 111111$ | (4) | (5) | (6) | (7) |
|---|---------------|---------------|--|---------------|---------------|---------------|---------------|
| Dependent Variable: | Does | Tends for | Engages in | Hours | Hours | Hours in | Total Child |
| Dependent variable. | Household | Siblings or | Farm Labor | Doing | Tending for | Farm Labor | Labor Hours |
| | Chores | Sick | raiiii Lauui | Household | Siblings or | raiiii Lauui | Laudi Hours |
| | Chores | Members | | Chores | Sick | | |
| | | Members | | CHOICS | Members | | |
| Instruments | Rain Shocks | Rain Shocks | Rain Shocks | Rain Shocks | Rain Shocks | Rain Shocks | Rain Shocks |
| Instruments: | | | | | | | |
| | for Harvests | for Harvests | for Harvests | for Harvests | for Harvests | for Harvests | for Harvests |
| | for In Utero, | for In Utero, | for In Utero, | for In Utero, | for In Utero, | for In Utero, | for In Utero, |
| | Age 0 , or | Age 0 , or | Age 0 , or | Age 0 , or | Age 0, or Age | Age 0 , or | Age 0 , or |
| | Age 1 | Age 1 | Age 1 | Age 1 | 1 | Age 1 | Age 1 |
| | | | | | | | |
| Ability (Raven's age | 0.368 | -0.296** | -0.279* | -2.750* | -1.038* | -5.080*** | -11.529*** |
| adjusted z-score) | [0.540] | [0.122] | [0.143] | [1.589] | [0.580] | [1.847] | [3.240] |
| Male | -0.290*** | -0.097*** | 0.138*** | -2.788*** | -0.320*** | 0.980*** | -1.908*** |
| | [0.061] | [0.026] | [0.026] | [0.325] | [0.118] | [0.308] | [0.692] |
| Household Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Age Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| First Stage F-Statistic of Excluded Instruments | 9.69 | 9.69 | 9.69 | 9.69 | 9.69 | 9.69 | 9.69 |
| Observations | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 |

Notes: Robust standard errors in brackets, clustered at village level. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include household fixed effects and child age and gender dummies. Regression sample includes 2862 children ages 5-15 from households with multiple siblings in this age range with differing enrollment outcomes. Ability is measured using the Raven's Colored Progressive Matrices and normed by age (z-score). All dependent variables are defined in the notes to Table 1. The 3 instruments measure if rainfall for the harvest for the in utero, age 0, or age 1 years in the child's village are 1 standard deviation below the historical average rainfall for the 15-year period before any of the survey children were born (1976-1991). Data source: Nahouri Cash Transfer Pilot Program (NCTPP) 2008 household survey data and rainfall data from Burkina Faso and Ghana Meteorological Services.

Table 6: Placebo Household Fixed Effects IV Regressions Estimating Relationship between Ability, Schooling, and Child Labor, Treating Ability as Endogenous, But Using Rain Shocks Years Before Child In Utero

| | (1) | (2) | (3) | (4) | (5) | (6) | (7) |
|-------------------------------|------------|-------------|-------------|---------------------------------------|--------------|----------|----------|
| Panel A: Schooling IV Regress | | () | () | · · · · · · · · · · · · · · · · · · · | X / | | · / |
| Dependent Variable: | Current | Ever | Grade | Late Start | | | |
| | Enrollment | Enrolled | Progression | | | | |
| Ability (Raven's age adjusted | -0.401 | -0.651 | -0.397 | 0.478 | | | |
| z-score) | [0.736] | [1.044] | [0.819] | [0.916] | | | |
| Household Fixed Effects? | Yes | Yes | Yes | Yes | | | |
| Age & Gender Fixed Effects? | Yes | Yes | Yes | Yes | | | |
| First Stage F-Statistic of | 0.968 | 0.968 | 0.968 | 0.968 | | | |
| Excluded Instruments | | | | | | | |
| Observations | 2,862 | 2,862 | 2,714 | 2,862 | | | |
| Panel B. Child Labor IV Regre | ssions | | | | | | |
| Dependent Variable: | Does | Tends for | Engages in | Hours Doing | Hours | Hours in | Total |
| | Household | Siblings or | Farm Labor | Household | Tending for | Farm | Child |
| | Chores | Sick | | Chores | Siblings or | Labor | Labor |
| | | Members | | | Sick Members | | Hours |
| Ability (Raven's age adjusted | 0.200 | 0.582 | -0.336 | 4.550 | 5.392 | -2.285 | 9.500 |
| z-score) | [0.527] | [0.641] | [0.446] | [6.349] | [5.233] | [2.735] | [12.701] |
| Household Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Age & Gender Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| First Stage F-Statistic of | 0.968 | 0.968 | 0.968 | 0.968 | 0.968 | 0.968 | 0.968 |
| Excluded Instruments | | | | | | | |
| Observations | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 |

Notes: Robust standard errors in brackets, clustered at village level. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include household fixed effects and child age and gender dummies. Regression sample includes 2862 children ages 5-15 from households with multiple siblings in this age range with differing enrollment outcomes. Ability is measured using the Raven's Colored Progressive Matrices and normed by age (z-score). All dependent variables are defined in the notes to Table 1. The placebo instrument is if rainfall for the harvest 2 *years prior* to birth in the child's village is 1 standard deviation below the historical average rainfall for the 15-year period before any of the survey children were born (1976-1991). Data source: Nahouri Cash Transfer Pilot Program 2008 household survey data and rainfall data from Burkina Faso and Ghana Meteorological Services.

Table 7: Robustness Check: Household Fixed Effects IV Regressions Estimating Relationship Between Ability, Schooling and Child Labor, Treating Ability as Endogenous, Alternative Measures of Ability

| | (1) | (2) | (3) | (4) | (5) | (6) | (7) |
|-------------------------------|------------|---------------|-------------|-------------|--------------|-----------|----------|
| Panel A: Schooling IV Regress | | | (-) | () | (-) | (-) | () |
| Dependent Variable: | Current | Ever Enrolled | Grade | Late Start | | | |
| - | Enrollment | | Progression | | | | |
| Raven's Raw Score | 0.264*** | 0.241*** | 0.171*** | -0.550** | | | |
| | [0.063] | [0.037] | [0.038] | [0.232] | | | |
| Household Fixed Effects? | Yes | Yes | Yes | Yes | | | |
| Age & Gender Fixed Effects? | Yes | Yes | Yes | Yes | | | |
| First Stage F-Statistic of | 12.87 | 12.87 | 14.16 | 12.87 | | | |
| Excluded Instruments | | | | | | | |
| Observations | 2,862 | 2,862 | 2714 | 2,862 | | | |
| Panel B: Child Labor IV Regr | essions | | | | | | |
| Dependent Variable: | Does | Tends for | Engages in | Hours Doing | Hours | Hours in | Total |
| | Household | Siblings or | Farm Labor | Household | Tending for | Farm | Child |
| | Chores | Sick | | Chores | Siblings or | Labor | Labor |
| | | Members | | | Sick Members | | Hours |
| Raven's Raw Score | 0.115 | -0.092*** | -0.056* | -0.920 | -0.272 | -1.479*** | -3.533** |
| | [0.166] | [0.025] | [0.030] | [0.780] | [0.179] | [0.487] | [1.524] |
| Household Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Age & Gender Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| First Stage F-Statistic of | 12.87 | 12.87 | 12.87 | 12.87 | 12.87 | 12.87 | 12.87 |
| Excluded Instruments | | | | | | | |
| Observations | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 |

Notes: Robust standard errors in brackets, clustered at village level. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include household fixed effects and child age and gender dummies. Regression sample includes 2862 children ages 5-15 from households with multiple siblings in this age range with differing enrollment outcomes. Ability is measured using the Raven's Colored Progressive Matrices. All dependent variables are defined in the notes to Table 1. The 3 instruments measure if rainfall for the harvest for the in utero, age 0, or age 1 years in the child's village are 1 standard deviation below the historical average rainfall for the 15-year period before any of the survey children were born (1976-1991). Data source: Nahouri Cash Transfer Pilot Program 2008 household survey data and rainfall data from Burkina Faso and Ghana Meteorological Services.

Table 8: Characteristics of Households Having A Child During Rain Shock Periods

| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | (9) |
|--------------------|--------------|---------|----------|-----------|-----------|-----------|---------|-------------|-----------|
| Dependent | Per Capita | Log | Either | Marital | Age of | Household | Number | Number of | Household |
| Variable: | Household | Assets | Parent | Status | Household | Head is a | of | Children of | Size |
| | Expenditures | | Ever | Household | Head | Farmer | Wives | Household | |
| | | | Attended | Head | | | | Head | |
| | | | School | | | | | | |
| Having A Child | 1,420.659 | -0.260 | -0.056 | -0.078 | -2.766 | -0.063 | -0.262 | -0.710 | -1.274 |
| During Rain | [9,035.707] | [0.302] | [0.048] | [0.107] | [2.121] | [0.067] | [0.199] | [0.441] | [1.015] |
| Shock Periods | _ | _ | _ | _ | _ | _ | _ | _ | _ |
| Observations | 893 | 889 | 893 | 893 | 798 | 893 | 893 | 893 | 893 |

Notes: Robust standard errors in brackets, clustered at village level. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions at household level. Regression sample includes 893 households with multiple children ages 5-15 with differing enrollment outcomes. Independent variable for each regression measures if a household experienced for any of their children a negative rainfall shock where the rainfall for the harvest for the in utero, age 0, or age 1 years in the child's village are 1 standard deviation below the historical average rainfall for the 15-year period before any of the survey children were born (1976-1991). Data source: Nahouri Cash Transfer Pilot Program (NCTPP) 2008 household survey data and rainfall data from Burkina Faso and Ghana Meteorological Services.

Appendix Table 1: Household Fixed Effects IV Regressions Estimating Relationship Between Ability and Schooling,
Treating Ability as Endogenous, Using Single Year Instruments

| | (1) | (2) | (3) | (4) |
|-------------------------------|----------------------------|------------------------|-------------------|------------|
| Dependent Variables: | Current Enrollment | Ever Enrolled | Grade Progression | Late Start |
| Panel A: Rain Shock for Harve | st for Child's In Utero Ye | ar as Instrument for . | Ability | |
| Ability (Raven's age adjusted | 1.437*** | 1.123*** | 0.914*** | -3.531*** |
| z-score) | [0.285] | [0.226] | [0.211] | [0.736] |
| Panel B: Rain Shock for Harve | st for Child's Age 0 Year | as Instrument for Ab | ility | |
| Ability (Raven's age adjusted | 1.036*** | 0.797*** | 1.121** | -2.470* |
| z-score) | [0.319] | [0.262] | [0.550] | [1.346] |
| Panel C: Rain Shock for Harve | st for Child's Age 1 Year | as Instrument for Ab | ility | |
| Ability (Raven's age adjusted | 0.446*** | 0.601*** | 0.224 | -0.435 |
| z-score) | [0.136] | [0.129] | [0.193] | [0.689] |
| Household Fixed Effects? | Yes | Yes | Yes | Yes |
| Age & Gender Fixed Effects? | Yes | Yes | Yes | Yes |
| Observations | 2,862 | 2,862 | 2,714 | 2,862 |

Notes: Robust standard errors in brackets, clustered at village level. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include household fixed effects and child age and gender dummies. Regression sample includes 2862 children ages 5-15 from households with multiple siblings in this age range with differing enrollment outcomes. Ability is measured using the Raven's Colored Progressive Matrices and normed by age (z-score). All dependent variables are defined in the notes to Table 1. The 3 instruments measure if rainfall for the harvest for the in utero, age 0, or age 1 years in the child's village are 1 standard deviation below the historical average rainfall for the 15-year period before any of the survey children were born (1976-1991). Data source: Nahouri Cash Transfer Pilot Program (NCTPP) 2008 household survey data and rainfall data from Burkina Faso and Ghana Meteorological Services.

Appendix Table 2: Household Fixed Effects IV Regressions Estimating Relationship Between Ability and Child Labor,
Treating Ability as Endogenous, Using Single Year Instruments

| | (1) | (2) | (3) | (4) | (5) | (6) | (7) |
|-----------------------------|------------------|---------------------|-----------------|-------------|-----------------|----------|-------------|
| Dependent Variable: | Does | Tends for | Engages in | Hours Doing | Hours Tending | Hours in | Total Child |
| | Household | Siblings or | Farm | Household | for Siblings or | Farm | Labor Hours |
| | Chores | Sick Members | Labor | Chores | Sick Members | Labor | |
| Panel A: Rain Shock for Ho | arvest for Child | l's In Utero Year d | is Instrument f | for Ability | | | |
| Ability (Raven's age | 1.254*** | -0.242** | 0.504*** | -0.417 | -1.052** | -0.228 | -2.143 |
| adjusted z-score) | [0.262] | [0.096] | [0.140] | [1.223] | [0.481] | [0.557] | [2.031] |
| Panel B: Rain Shock for Ho | arvest for Child | l's Age 0 Year as 1 | Instrument for | Ability | | | |
| Ability (Raven's age | 0.216 | -0.350*** | 0.327 | -5.364 | 0.216 | -4.108* | -13.660 |
| adjusted z-score) | [0.733] | [0.114] | [0.594] | [8.774] | [0.516] | [2.366] | [15.246] |
| Panel C: Rain Shock for Ho | arvest for Child | d's Age 1 Year as I | Instrument for | Ability | | | |
| Ability (Raven's age | 0.030 | -0.299 | -0.884 | -2.725* | -1.550 | -7.678 | -14.898* |
| adjusted z-score) | [0.294] | [0.255] | [0.851] | [1.610] | [1.141] | [5.578] | [8.866] |
| Household Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Age & Gender Fixed Effects? | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Observations | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 | 2,862 |

Notes: Robust standard errors in brackets, clustered at village level. * significant at 10%; ** significant at 5%; *** significant at 1%. All regressions include household fixed effects and child age and gender dummies. Regression sample includes 2862 children ages 5-15 from households with multiple siblings in this age range with differing enrollment outcomes. Ability is measured using the Raven's CPM and normed by age (z-score). All dependent variables are defined in the notes to Table 1. The 3 instruments measure if rainfall for the harvest for the in utero, age 0, or age 1 years in the child's village are 1 standard deviation below the historical average rainfall for the 15-year period before any of the survey children were born (1976-1991). Data source: Nahouri Cash Transfer Pilot Program (NCTPP) 2008 household survey data and rainfall data from Burkina Faso and Ghana Meteorological Services.